\documentclass[10pt,draft]{amsart}
\usepackage[cp1251]{inputenc}
\usepackage[english,russian]{babel}
\usepackage{amsmath}
\usepackage{amssymb}
\usepackage{amsfonts}
\usepackage{eucal}

\def\udcs{515.57} %Here the author places classificators of the paper according to Russian classification system
\def\mscs{08C15, 54H99, 03C60, 08A30} %Here the author places  classificators according to the AMS classification list.
\setcounter{page}{1}

\newtheorem{thm}{Theorem}
\newtheorem{lem}[thm]{Lemma}
\newtheorem{prop}[thm]{Proposition}
\newtheorem{cor}[thm]{Corollary}
\theoremstyle{definition}
\newtheorem{definition}{Definition}
\newtheorem{remark}{Remark}
\newtheorem{problem}{Problem}

\theoremstyle{definition}
\newtheorem{claim}{Claim}
\newenvironment{scproof}
{\begin{proof}[Proof of Claim.]}
{\end{proof}}

\renewcommand{\labelenumi}{{\textrm{(\roman{enumi})}}}
\newcommand{\Case}[1]{\smallskip\noindent{\emph{Case~{#1}}}}
\numberwithin{equation}{section}

\def\logo{{\bf\huge S\raisebox{0.2ex}{\hspace{0.55ex}\raisebox{0.05ex}e\hspace{-1.65ex}$\bigcirc$}MR}}

\def\semrtop
     {
  \vbox{
     \noindent\logo
     \hspace{80mm}\raisebox{1ex}{ISSN 1813-3304 }

     \vspace{5mm}

     \begin{center}
     {\huge СИБИРСКИЕ \ ЭЛЕКТРОННЫЕ} \\[2mm]  %Headings in Russian
     {\huge МАТЕМАТИЧЕСКИЕ ИЗВЕСТИЯ} \\[2mm]
     {\large Siberian Electronic Mathematical Reports} \\[1mm]
     {\LARGE\tt{http://semr.math.nsc.ru}}\\[0.5mm]
%     {\small Институт математики им. С.Л.Соболева СО РАН}\\[-1mm]
%     {\small Sobolev Institute of Mathematics SB RAS}
     \end{center}
     \vspace{-3mm}
     \noindent
     \begin{tabular}{c}
     \hphantom{aaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaa} \\
     \hline\hline
     \end{tabular}

     \vspace{1mm}
     {\flushleft\it Том 20, стр. 144--144 (2023) \hspace{\fill}{\rm\small УДК \udcs}} %Volume, pages in Russian,
                                                                                    %filled by Editorial board
     \newline
         {\rm\small DOI~10.17377/semi.2022.12.xxx}\hphantom{aaaaaaaaaaaaaaaaaaaaaaaa}{\rm\small MSC\ \ \mscs }
  }%\vbox
}

\begin{document}
\begin{thebibliography}{99}

\bibitem{AD1}
M.\,E. Adams, W. Dziobiak,
$Q$-\emph{universal quasivarieties of algebras},
Proc.\ Amer.\ Math.\ Soc. \textbf{120}, no. 4 (1994), 1053--1059.

\bibitem{BNSZ}
A. Basheyeva, A. Nurakunov, M. Schwidefsky, and A. Zamojska-Dzienio,
\emph{Lattices of subclasses.} III,
Siberian Electronic Mathematical Reports \textbf{14} (2017), 252--263;\\
available at {\tt http://semr.math.nsc.ru/v14/p252-263.pdf}.

\bibitem{CD}
D.\,M. Clark, B.\,A. Davey,
\emph{Natural Dualities for the Working Algebraist},
Cambridge University Press, Cambridge, 1998.

\bibitem{CDF}
D.\,M. Clark, B.\,A. Davey, R.\,S. Freese, and M. Jackson,
\emph{Standard topological algebras: syntactic and principal congruences and profiniteness},
Algebra Universalis \textbf{52}, no.~2-3 (2005), 343--376.

\bibitem{CDH}
D.\,M. Clark, B.\,A. Davey, M. Haviar, J.\,G. Pitkethly, and M.\,R. Talukder,
\emph{Standard topological quasi-varieties},
Houston J.\ Math. \textbf{29}, no.~4 (2003), 859--887.

\bibitem{CDJP}
D.\,M. Clark, B.\,A. Davey, M.\,G. Jackson, and J.\,G. Pitkethly,
\emph{The axiomatizability of topological prevarieties},
Adv.\ Math. \textbf{218}, no.~5 (2008), 1604--1653.

\bibitem{DJMM}
B.\,A. Davey, M. Jackson, M. Maroti, R.\,N. McKenzie,
\emph{Principal and syntactic congruences in congruence-distributive and congruence-permutable varieties},
J.\ Austral.\ Math.\ Soc. \textbf{85}, no.~1 (2008), 59--74. 

\bibitem{G}
V.\,A. Gorbunov,
\emph{Algebraic Theory of Quasivarieties},
Plenum, New York, 1998.

\bibitem{KNS1}
A.\,V. Kravchenko, A.\,M. Nurakunov, M.\,V. Schwidefsky,
\emph{Structure of quasivariety lattices}. I. \emph{Independent axiomatizability},
Algebra and Logic \textbf{57}, no.~6 (2018), 445--462.

\bibitem{KNS2}
A.\,V. Kravchenko, A.\,M. Nurakunov, M.\,V. Schwidefsky,
\emph{Structure of quasivariety lattices}. II. \emph{Undecidable problems},
Algebra and Logic \textbf{58}, no.~2 (2019), 123--136.

\bibitem{KNS3}
A.\,V. Kravchenko, A.\,M. Nurakunov, M.\,V. Schwidefsky,
\emph{Structure of quasivariety lattices}. III. \emph{Finitely partitionable bases},
Algebra and Logic \textbf{59}, no.~3 (2020), 222--229.

\bibitem{KNS4}
A.\,V. Kravchenko, A.\,M. Nurakunov, M.\,V. Schwidefsky,
\emph{Structure of quasivariety lattices}. IV. \emph{Nonstandard quasivarieties},
Siberian Math. J. \textbf{62}, no.~5 (2021), 850--858.

\bibitem{KNS-AU}
A.\,V. Kravchenko, A.\,M. Nurakunov, M.\,V. Schwidefsky,
with A.\,V. Kravchenko and A.\,M. Nurakunov \emph{On representation of finite lattices},
Algebra Universalis \textbf{80}, no.~1 (2019), paper no. 15.

\bibitem{KNS-IJAC}
A.\,V. Kravchenko, A.\,M. Nurakunov, M.\,V. Schwidefsky,
\emph{On the complexity of the lattices of subvarieties and congruences},
Internat.\ J.\ Algebra Comput. \textbf{30}, no.~8 (2020), 1609--1624.

\bibitem{DG-UA-1}
A.\,V. Kravchenko, A.\,M. Nurakunov, M.\,V. Schwidefsky,
\emph{Quasi-equational bases of differential groupoids and unary algebras},
Siberian Electronic Mathematical Reports \textbf{14} (2017), 1330--1337;
available at {\tt http://semr.math.nsc.ru/v14/p1330-1337.pdf}.

\bibitem{DG-UA-2}
A.\,V. Kravchenko, M.\,V. Schwidefsky,
\emph{On the complexity of variety lattices and congruence lattices.} II. \emph{Differential groupoids and unary algebras},
Siberian Electronic Mathematical Reports \textbf{17} (2020), 753--768;\\
available at {\tt http://semr.math.nsc.ru/v17/p753-768.pdf}

\bibitem{DG-UA-3}
A.\,V. Kravchenko, M.\,V. Schwidefsky,
\emph{On non-standard quasivarieties of differential group\-oids and unary algebras},
Siberian Electronic Mathematical Reports \textbf{19}, no.~2 (2022), 768--783;\\
available at {\tt http://semr.math.nsc.ru/v19/n2/p768-783.pdf}.

\bibitem{KS16}
V. Koubek, J. Sichler,
\emph{Universality of small lattice varieties},
Proc.\ Amer.\ Math.\ Soc. \textbf{91}, no. 1 (1984), 19--24.

\bibitem{KS18}
V. Koubek, J. Sichler,
\emph{On almost universal varieties of modular lattices},
Algebra Universalis \textbf{45}, no. 2--3 (2001), 191--210.

\bibitem{KS19}
V. Koubek, J. Sichler,
\emph{Almost} ${f}{f}$-\emph{universal and} $Q$-\emph{universal varieties of modular} $0$-\emph{lattices},
College Math.\ J. \textbf{101}, no. 2 (2004), 161--182.

\bibitem{KS21}
V. Koubek, J. Sichler,
\emph{Finitely generated almost universal varieties of} $0$-\emph{lattices},
Comment.\ Math.\ Univ.\ Carolin. \textbf{46}, no. 2 (2005), 304--325.

\bibitem{KS2}
V. Koubek, J. Sichler,
\emph{On synchronized relatively full embeddings and} $Q$-\emph{universality},
Cahiers de Topologie et G\'{e}om\'{e}trie Diff\'{e}rentielle Cat\'{e}goriques \textbf{XLIX}, no. 4 (2008), 289--306.

\bibitem{KS1}
V. Koubek, J. Sichler,
\emph{Almost} $ff$-\emph{universality implies} $Q$-\emph{universality},
Appl.\ Categor.\ Struct. \textbf{17}, no. 5 (2009), 419--434.

\bibitem{NS}
A.\,M. Nurakunov, M.\,V. Schwidefsky,
\emph{Profinite locally finite quasivarieties},
submitted for publication in 2022.

\bibitem{NSt}
A.\,M. Nurakunov, M.\,M. Stronkowski,
\emph{Profiniteness in finitely generated varieties is undecidable},
J.\ Symb.\ Logic \textbf{83}, no.~4 (2018), 1566--1578.

\bibitem{PT}
A. Pultr and V. Trnkov\'{a},
\emph{Combinatorial, Algebraic and Topological Representations of Groups, Semigroups and Categories}, 
North Holland, Amsterdam, 1980.

\bibitem{RS}
A.\,B. Romanowska, J.\,D.\,H. Smith,
\emph{Modes},
World Scientific, Singapore, 2002.

\bibitem{S15}
M.\,V. Schwidefsky,
\emph{On complexity of quasivariety lattices},
Algebra and Logic \textbf{54}, no.~3 (2015), 245--257.

\bibitem{S20}
M.\,V. Schwidefsky,
\emph{On sufficient conditions for $Q$-universality},
Siberian Electronic Mathematical Reports \textbf{17} (2020), 1043--1051;\\
available at {\tt http://semr.math.nsc.ru/v17/p1043-1051.pdf}.

\bibitem{B-I}
M.\,V. Schwidefsky,
\emph{Existence of independent quasi-equational bases},
Algebra and Logic \textbf{58}, no. 6 (2020), 514--537.

\bibitem{B-II}
M.\,V. Schwidefsky,
\emph{On the existence of independent quasi-equational bases}. II,
to appear in Algebra and Logic. %\textbf{58}, no. 6 (2023), 514-537.

\bibitem{SZ}
M.\,V. Schwidefsky, A. Zamojska-Dzienio
\emph{Lattices of subclasses.} II,
Internat.\ J.\ Algebra Comput. \textbf{24}, no.~8 (2014), 1099--1126.

\end{thebibliography}

\renewcommand{\refname}{References}
\renewcommand{\proofname}{Proof.}
\renewcommand{\figurename}{Figure}
\thispagestyle{empty}

\title[The complexity of quasivariety lattices. II]{\sc The complexity of quasivariety lattices. II}
\author{M.~V. Schwidefsky}
\address{Marina Vladimirovna Schwidefsky
\newline\hphantom{iii} %${}^{\rm (a)}$ 
Novosibirsk State University,
\newline\hphantom{iii} Pirogova str. 1,
\newline\hphantom{iii} 
630090, Novosibirsk, Russia}%
\email{m.schwidefsky@g.nsu.ru}%

\thanks{\sc Schwidefsky, M.~V., The complexity of quasivariety lattices. II}
\thanks{\copyright \ 2023 Schwidefsky, M.~V.}
\thanks{The research was carried out under the support of the Russian Science Foundation, project no. 22-21-00104.}
\thanks{\it Received  March 20, 2022, published  ????.}%


\semrtop \vspace{1cm}
\maketitle {\small
\begin{quote}
\noindent{\sc Abstract. } We prove that if a quasivariety $\mathbf{K}$ contains a finite $\mathrm{B}^\ast$-class relative to some subquasivariety and some variety
possessing some additional property, then $\mathbf{K}$ contains continuum many $Q$-universal non-profinite subquasivarieties having an independent quasi-equational basis as well as continuum many
$Q$-universal non-profinite subquasivarieties having no such basis.

\medskip

\noindent{\bf Keywords:} inverse limit, quasi-equational basis, quasivariety, profinite structure, profinite quasivariety.
 \end{quote}
}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%



\section{Introduction}\label{S:Intro}

\noindent
Under certain sufficient condition, quasivarieties and their subquasivariety lattices turn out highly complicated structure~--- from the structural, syntactical, algorithmic, as well as topological point of view. First condition of this type (which ensures the existence of a countable subclass possessing certain properties) were found by M.\,E. Adams and W. Dziobiak \cite{AD1}; a bit later, similar conditions were proposed by V.\,A. Gorbunov \cite{G}. Relatively recently,
A.\,V. Kravchenko \emph{et al.} \cite{KNS1} suggested another sufficient condition, the existence of so-called $\mathrm{B}$-class in a quasivariety,
which implies that the quasivariety has complicated inner structure, see \cite{KNS1}--\cite{KNS4} and \cite{KNS-AU,KNS-IJAC}. However, it turned out that this condition was very strong as
very similar complexity results hold for certain concrete quasivarieties of algebraic structures possessing no $\mathrm{B}$-class,
see \cite{DG-UA-1}--\cite{DG-UA-3}, \cite{BNSZ,SZ}.

In \cite{B-I}, a weaker sufficient condition, the existence of a $\mathrm{B}^\ast$-class in a quasivariety, was introduced, see Definition \ref{D:B}.
It was established in \cite{B-I,B-II} that quasivarieties having $\mathrm{B}^\ast$-classes possess many of the properties of quasivarieties with
$\mathrm{B}$-classes.

The present paper continues \cite{S15}. We establish here that if a quasivariety $\mathbf{K}$ contains a finite $\mathrm{B}^\ast$-class relative to some subquasivariety and some variety possessing some additional property which we call $(\mathrm{B}^\ast)$,
then $\mathbf{K}$ contains continuum many $Q$-universal non-profinite subquasivarieties having an independent quasi-equational basis as well as continuum many $Q$-universal non-profinite subquasivarieties having no such basis, see Theorems \ref{T:1} and \ref{T:2}.
The main results of \cite{KNS4} and \cite{DG-UA-3} turn out to be corollaries of these two theorems.
We note that the notion of profiniteness was intensively studied for [quasi]varieties by D.\,M. Clark \emph{et al.} in \cite{CD}--\cite{DJMM}, see also A.\,M. Nurakunov
and M.\,M. Stronkowski \cite{NSt}.

In Section \ref{A}, we present some applications of our main results to concrete quasivarieties of algebraic structures.

\section{Main definitions and auxiliary results}

\noindent
This section contains all the necessary definitions and results established earlier that we use in the present paper.
For all the definitions and notation which are note defined here, we refer to \cite{KNS4,B-II}.

\subsection{$\mathrm{B}^\ast$-classes}\label{S:ast}
Here, we present our central definition, the definition of $\mathrm{B}^\ast$-class, as it appeared in \cite{B-I}.

We denote by $\mathcal{P}_{fin}(\omega)$ the set of finite subsets of the set $\omega$ of natural numbers.
Let $J\subseteq\omega$ be an infinite set. We may assume that $J=\{j_n\mid n<\omega\}$, where  $j_m<j_n$ if and only if $m<n$.
For a set $T\subseteq\omega$, we put $J(T)=\{j_n\in J\mid n\in T\}$.

\begin{definition}\label{D:B}
Let $\mathbf{M}\subseteq\mathbf{K}(\sigma)$ be a quasivariety of a finite similarity type $\sigma$
and let $\mathbf{V}\subseteq\mathbf{K}(\sigma)$ be a nonempty homomorphically closed class.
A class $\mathbf{A}=\{\mathcal{A}_F\mid F\in\mathcal{P}_{fin}(\omega)\}\subseteq\mathbf{M}$ is called a $\mathrm{B}^\ast$-\emph{class with respect to} $\mathbf{M}$ \emph{and} $\mathbf{V}$, if $\mathbf{A}$ satisfies the following conditions:
\begin{enumerate}
\item[$(\mathrm{B}_0)$]
for each nonempty $F\in\mathcal{P}_{fin}(\omega)$, the structure $\mathcal{A}_F$ is finitely presented in $\mathbf{M}$;
$\mathcal{A}_\varnothing$ is a trivial structure;
\item[$(\mathrm{B}_1)$]
if $F=G\cup H$ in $\mathcal{P}_{fin}(\omega)$
then $\mathcal{A}_F\in\mathbf{Q}(\mathcal{A}_G,\mathcal{A}_H)$;
\item[$(\mathrm{B}^\ast_2)$]
for each $F$, $G\in\mathcal{P}_{fin}(\omega)$, if $F\ne\varnothing$ and
$\mathcal{A}_F\in\mathbf{Q}(\mathcal{A}_G,\mathbf{V})$ then $F=G$;
\item[$(\mathrm{B}^\ast_3)$]
for every $F\in\mathcal{P}_{fin}(\omega)$ and every $i<\omega$, if
$f\in\operatorname{Hom}(\mathcal{A}_F,\mathcal{A}_{\{i\}})$ then either $f(\mathcal{A}_F)\in\mathbf{V}$ or $i\in F$;
\item[$(\mathrm{B}^\ast_4)$]
for all $F\in\mathcal{P}_{fin}(\omega)$,
$\bigl(\mathbf{H}(\mathcal{A}_F)\cap\mathbf{M}\bigr){\setminus}\mathbf{V}\subseteq\mathbf{A}$.
\end{enumerate}
If $\mathcal{A}_F$ is a finite structure for every nonempty set $F\in\mathcal{P}_{fin}(\omega)$, then we call
$\mathbf{A}$ a \emph{finite} $\mathrm{B}^\ast$-\emph{class with respect to} $\mathbf{M}$ \emph{and} $\mathbf{V}$.
\end{definition}

\noindent
Definition \ref{D:B} was inspired by M.\,E. Adams and W. Dziobiak \cite{AD1} as well as by
A.\,V. Kravchenko \emph{et al.} \cite[Definition 2.1]{KNS1}.
The next statement is a straightforward corollary of the definition of $\mathrm{B}$-class given in \cite{KNS1}.

\begin{cor}\label{C:b-b}
If $\mathbf{A}$ is a $\mathrm{B}$-class with respect to some quasivariety $\mathbf{M}$ then
$\mathbf{A}$ is a $\mathrm{B}^\ast$-class
with respect to $\mathbf{M}$ and the trivial variety $\mathbf{T}$.
\end{cor}

\subsection{Properties $(\mathrm{B}^\ast_\Theta)$ and $(\mathrm{B}^\ast)$}\label{ss:th}
Let $\mathbf{M}\subseteq\mathbf{K}(\sigma)$ be a quasivariety of a finite similarity type $\sigma$,
let $\mathbf{V}\subseteq\mathbf{K}(\sigma)$ be a nonempty homomorphically closed class, and let
a class $\mathbf{A}=\{\mathcal{A}_F\mid F\in\mathcal{P}_{fin}(\omega)\}\subseteq\mathbf{M}$ be a $\mathrm{B}^\ast$-class with respect to $\mathbf{M}$ and $\mathbf{V}$.
Consider a nonempty set $F\in\mathcal{P}_{fin}(\omega)$. According to condition 
$(\mathrm{B}_0)$ from Definition \ref{D:B}, there are a finite set $X_F$
of variables and a finite set $\Delta_F$ of atomic formulas with free variables belonging to $X_F$ such that
$\mathcal{A}_F\cong\mathcal{F}_{\mathbf{M}}(X_F,\Delta_F)$.
Let $\gamma_F\colon X_F\to A_F$ be the corresponding interpretation of the variables from $X_F$.

Furthermore, let $G\subseteq F$.
By Lemma \ref{L:B}(ii) below, there is a homomorphism from $\mathcal{A}_F$ onto $\mathcal{A}_G$.
We fix one such homomorphism and denote it by $f^F_G$.
By \cite[Lemma 1.2]{KNS1}, there is a finite set $\Delta^F_G$ of atomic formulas and an interpretation $\gamma^F_G\colon X_F\to A_G$
such that $\mathcal{A}_G\cong\mathcal{F}_{\mathbf{M}}(X_F,\Delta^F_G)$ with respect to $\gamma^F_G$.
In this case, we have also $\Delta^F_G\models_{\mathbf{M}}\Delta_F$ and
$f^F_G\bigl(\gamma_F(x)\bigr)=\gamma^F_G(x)$ for all $x\in X_F$.
We assume that $\Delta^F_F=\Delta_F$ and $\gamma^F_F=\gamma_F$.
We put
\begin{align*}
&\Theta=\bigl(\Delta(\Theta),\Gamma(\Theta),\Omega(\Theta)\bigr),\quad\text{where}\\
&\Delta(\Theta)=\bigl\{(X_F,\Delta_F,\gamma_F)\mid F\in\mathcal{P}_{fin}(\omega){\setminus}\{\varnothing\}\bigr\};\\
&\Gamma(\Theta)=\{\gamma^F_G\mid F,G\in\mathcal{P}_{fin}(\omega){\setminus}\{\varnothing\},\ G\subseteq F\};\\
&\Omega(\Theta)=\{f^F_G\mid F,G\in\mathcal{P}_{fin}(\omega){\setminus}\{\varnothing\},\ G\subseteq F\}.
\end{align*}
We consider also the following condition on the class $\mathbf{A}$ which depends on $\Theta$:

\begin{enumerate}
\item[$(\mathrm{B}^\ast_\Theta)$]
for sets $F$, $G\in\mathcal{P}_{fin}(\omega)$ such that $\varnothing\ne G\subseteq F$,
for a structure $\mathcal{B}\in\mathbf{V}$ and a homomorphism
$f\in\operatorname{Hom}(\mathcal{A}_F,\mathcal{B})$,
there is a homomorphism $g\in\operatorname{Hom}(\mathcal{A}_G,\mathcal{B})$ such that
$g\bigl(\gamma^F_G(x)\bigr)=f\bigl(\gamma_F(x)\bigr)$ for all $x\in X_F$.
\end{enumerate}


\noindent
We also consider the following condition on the class $\mathbf{A}$:

\begin{enumerate}
\item[$(\mathrm{B}^\ast)$]
for sets $F$, $G\in\mathcal{P}_{fin}(\omega)$ such that $\varnothing\ne G\subseteq F$,
for a structure $\mathcal{B}\in\mathbf{V}$ and homomorphisms
$f\in\operatorname{Hom}(\mathcal{A}_F,\mathcal{B})$ and $g\in\operatorname{Hom}(\mathcal{A}_F,\mathcal{A}_G)$,
there is a homomorphism $h\in\operatorname{Hom}(\mathcal{A}_G,\mathcal{B})$ such that $f=hg$.
\end{enumerate}

\noindent
Properties $(\mathrm{B}^\ast_\Theta)$ and $(\mathrm{B}^\ast)$ were introduced and considered in \cite{B-I}.

\subsection{Auxiliary results on $\mathrm{B}^\ast$-classes}
We present here some results from \cite{B-I,B-II} which we need.

\begin{lem}\label{L:B}
\cite[Lemma 1.3]{B-I}
Let $\mathbf{M}\subseteq\mathbf{K}(\sigma)$ be a quasivariety
of a finite similarity type $\sigma$,
let $\mathbf{V}\subseteq\mathbf{K}(\sigma)$ be a nonempty homomorphically closed class,
and let a class $\mathbf{A}=\{\mathcal{A}_F\mid F\in\mathcal{P}_{fin}(\omega)\}\subseteq\mathbf{M}$ satisfy conditions $(\mathrm{B}_0)$ and $(\mathrm{B}^\ast_2)$ with respect to $\mathbf{M}$ and $\mathbf{V}$.
The following statements hold.
\begin{enumerate}
\item
If $\mathcal{A}_F\in\mathbf{V}$ for some $F\in\mathcal{P}_{fin}(\omega)$ then $F=\varnothing$.
\item
Let $\mathbf{A}$ satisfy in addition $(\mathrm{B}_1)$ and $(\mathrm{B}^\ast_4)$ with respect to $\mathbf{M}$ and $\mathbf{V}$.
If $G\subseteq F\in\mathcal{P}_{fin}(\omega)$ then $\mathcal{A}_G\in\mathbf{H}(\mathcal{A}_F)$.
\item
Let $\mathbf{A}$ be a $\mathrm{B}^\ast$-class with respect to $\mathbf{M}$ and $\mathbf{V}$.
If $f\in\operatorname{Hom}(\mathcal{A}_F,\mathcal{A}_G)$ for some $F$, $G\in\mathcal{P}_{fin}(\omega)$ then either $f(\mathcal{A}_F)\in\mathbf{V}$ or $G\subseteq F$ and $f(\mathcal{A}_F)\cong\mathcal{A}_G$.
\end{enumerate}
\end{lem}

\noindent
In the context of Definition \ref{D:B}, let $\mathbf{M}^\ast=(\mathbf{M}{\setminus}\mathbf{V})\cup\{\mathcal{E}\}$,
where $\mathcal{E}$ denotes the trivial structure.

\begin{lem}\label{L:B2}
\cite[Lemma 1.4]{B-I}
Let $\mathbf{M}\subseteq\mathbf{K}(\sigma)$ be a quasivariety, let
$\mathbf{V}\subseteq\mathbf{K}(\sigma)$ be a nonempty homomorphically closed class, let
a class $\mathbf{A}=\{\mathcal{A}_F\mid F\in\mathcal{P}_{fin}(\omega)\}\subseteq\mathbf{M}$
be a $\mathrm{B}^\ast$-class with respect to $\mathbf{M}$ and $\mathbf{V}$, and let $i<\omega$.
If $1_{\mathcal{A}_{\{i\}}}$ has a lower cover in the poset $\operatorname{Con}_{\mathbf{M}^\ast}\mathcal{A}_{\{i\}}$,
then $\mathcal{A}_{\{i\}}$ is an $\mathbf{M}^\ast$-simple structure.
\end{lem}

\begin{remark}\label{R:1}
We notice that Lemma \ref{L:B2} applies when $\sigma$ is a finite similarity type and the structure
$\mathcal{A}_{\{i\}}$ is finite for all $i<\omega$.
In particular, if $\mathbf{A}$ is a finite $\mathrm{B}^\ast$-class with respect to $\mathbf{M}$, $\mathbf{V}\subseteq\mathbf{K}(\sigma)$,
then $\mathbf{A}$ satisfies the following condition:
\begin{enumerate}
\item[$(\mathrm{B}^\ast_5)$]
for all $i<\omega$, $\mathcal{A}_{\{i\}}$ is an $\mathbf{M}^\ast$-simple structure.
\end{enumerate}
\end{remark}

\begin{lem}\label{L:H}
\cite[Lemma 2.3]{B-I}
Let $\mathbf{M}\subseteq\mathbf{K}(\sigma)$ be a quasivariety of a finite similarity type $\sigma$,
let $\mathbf{V}\subseteq\mathbf{K}(\sigma)$ be a variety, and let
a class $\mathbf{A}=\{\mathcal{A}_F\mid F\in\mathcal{P}_{fin}(\omega)\}\subseteq\mathbf{M}$ be a $\mathrm{B}^\ast$-class satisfying $(\mathrm{B}^\ast_\Theta)$ with respect to $\mathbf{M}$, $\mathbf{V}$, and some $\Theta$.
Then condition $(\mathrm{B}^\ast_5)$ is equivalent to each one of the following five conditions.
\begin{enumerate}
\item
For arbitrary sets $F,G\in\mathcal{P}_{fin}(\omega)$ such that $G\subseteq F$
and for arbitrary homomorphisms $f_0,f_1\in\operatorname{Hom}(\mathcal{A}_F,\mathcal{A}_G)$, if $f_0(\mathcal{A}_F)\notin\mathbf{V}$
then there is $g\in\operatorname{Hom}\bigl(f_0(\mathcal{A}_F),\mathcal{A}_G\bigr)$ such that $f_1=gf_0$;
\item
For arbitrary sets $F,G,H\in\mathcal{P}_{fin}(\omega)$ such that $F\subseteq G\subseteq H$
and for arbitrary homomorphisms $f\in\operatorname{Hom}(\mathcal{A}_H,\mathcal{A}_F)$ and $g\in\operatorname{Hom}(\mathcal{A}_H,\mathcal{A}_G)$ such that $g(\mathcal{A}_H)\notin\mathbf{V}$,
there is $h\in\operatorname{Hom}\bigl(g(\mathcal{A}_H),\mathcal{A}_F\bigr)$ such that $f=hg$.
\item
For arbitrary sets $F,G\in\mathcal{P}_{fin}(\omega)$ such that $G\subseteq F$
and for arbitrary homomorphisms $f_0,f_1\in\operatorname{Hom}(\mathcal{A}_F,\mathcal{A}_G)$ such that
$f_0$ is an onto homomorphism and $f_1(\mathcal{A}_F)\notin\mathbf{V}$,
there is an embedding $g\in\operatorname{Hom}(\mathcal{A}_G,\mathcal{A}_G)$ such that $f_1=gf_0$.
\item
If $F,G\in\mathcal{P}_{fin}(\omega)$ and $G\subseteq F$ then there exists a unique $\mathbf{M}$-congruence $\theta$ on $\mathcal{A}_F$ such that $\mathcal{A}_G\cong\mathcal{A}_F{\slash}\theta$.
\item
For an arbitrary set $F\in\mathcal{P}_{fin}(\omega)$, $\operatorname{Con}_{\mathbf{M}^\ast}\mathcal{A}_F\cong 2^F$, where
$2$ denotes a two-element lattice.
\end{enumerate}
\end{lem}

\noindent
The next statement is a corollary of Lemma \ref{L:H}.

\begin{cor}\label{C:B*}
\cite[Corollary 2.4]{B-I}
Let $\mathbf{M}\subseteq\mathbf{K}(\sigma)$ be a quasivariety of a finite similarity type $\sigma$,
let $\mathbf{V}\subseteq\mathbf{K}(\sigma)$ be a variety, and let
a class $\mathbf{A}=\{\mathcal{A}_F\mid F\in\mathcal{P}_{fin}(\omega)\}\subseteq\mathbf{M}$ be a $\mathrm{B}^\ast$-class satisfying $(\mathrm{B}^\ast_5)$ with respect to $\mathbf{M}$ and $\mathbf{V}$.
Then the class $\mathbf{A}$ satisfies $(\mathrm{B}^\ast)$ if and only if $\mathbf{A}$ satisfies
$(\mathrm{B}^\ast_\Theta)$ for some $\Theta$.
\end{cor}

\noindent
The next statements follows from the proof of \cite[Proposition 2.5]{B-I}.

\begin{prop}\label{P:cont}
Let a quasivariety $\mathbf{M}\subseteq\mathbf{K}(\sigma)$ contain a $\mathrm{B}^\ast$-class
$\mathbf{A}=\bigl\{\mathcal{A}_F\mid F\in\mathcal{P}_{fin}(\omega)\bigr\}$
satisfying $(\mathrm{B}^\ast)$
with respect to $\mathbf{M}$ and some variety $\mathbf{V}\subseteq\mathbf{K}(\sigma)$.
If $J\subseteq\omega$ is an infinite set then
\[
\mathbf{B}_J=\bigl\{\mathcal{A}_{J(F)}\mid F\in\mathcal{P}_{fin}(\omega)\bigr\}
\]
is a $\mathrm{B}^\ast$-class with respect to $\mathbf{M}$ and $\mathbf{V}$ which also satisfies $(\mathrm{B}^\ast)$.
\end{prop}

\begin{thm}\cite[Theorem 2.1]{B-II}\label{T:B}
Let a class $\mathbf{A}=\{\mathcal{A}_F\mid F\in\mathcal{P}_{fin}(\omega)\}$ of algebraic structures
of a finite similarity type $\sigma$
be a $\mathrm{B}^\ast$-class satisfying $(\mathrm{B}^\ast_5)$ and $(\mathrm{B}^\ast)$ with respect to a quasivariety $\mathbf{M}\subseteq\mathbf{K}(\sigma)$
and a variety $\mathbf{V}\subseteq\mathbf{K}(\sigma)$.
Then there are continuum many quasivarieties $\mathbf{K}\subseteq\mathbf{M}$ with the following properties:
\begin{enumerate}
\item
$\mathbf{K}$ has a finitely partitionable $\omega$-independent quasi-equational basis relative to $\mathbf{M}$;
\item
there are quasivarieties $\mathbf{K}_I\subseteq\mathbf{M}$, $I\in\mathfrak{I}$, such that
$\mathbf{K}=\bigcap_{I\in\mathfrak{I}}\mathbf{K}_I$ and the quasivariety $\mathbf{K}_I$
has no finitely partitionable $\omega$-independent quasi-equation\-al basis relative to $\mathbf{M}$, but has
an $\omega$-independent quasi-equational basis relative to $\mathbf{M}$ for all $I\in\mathfrak{I}$.
\end{enumerate}
If $\mathbf{A}$ is a finite $\mathrm{B}^\ast$-class then $\mathbf{K}$ above can be chosen $Q$-universal.
\end{thm}

\subsection{Profinite quasivarieties}
For a structure $\mathcal{A}\in\mathbf{K}(\sigma)$, we say that $\mathbb{A}=\langle\mathcal{A},\mathcal{T}\rangle$
is a \emph{topological structure},
if $\mathcal{T}$ is a topology on $A$ and all the basic operations of $\mathcal{A}$ are continuous and all the basic relations on $\mathcal{A}$
are closed with respect to $\mathcal{T}$. For a topological structure $\mathbb{A}$, we denote its algebraic reduct by $\mathcal{A}$ and its topology by $\mathcal{T}(\mathbb{A})$.
A topology $\mathcal{T}$ on a set $A$ is \emph{Boolean} if the topological space $\langle A,\mathcal{T}\rangle$ is Hausdorff, compact and has a base of clopen sets.
A topological structure $\mathbb{A}$ is \emph{Boolean} if $\mathcal{T}(\mathbb{A})$ is a Boolean topology.

A structure $\mathcal{A}\in\mathbf{K}(\sigma)$ is \emph{profinite} if it is isomorphic to an inverse limit of finite structures.
Profinite structures are naturally equipped with Boolean topologies which are in this case product topologies of discrete topologies on finite spaces.
A prevariety $\mathbf{K}$ is \emph{profinite} if each profinite structure belonging to $\mathbf{K}$ is \emph{profinite with respect to} $\mathbf{K}$;
that is, is isomorphic to an inverse limit of finite structures belonging to $\mathbf{K}$.

\begin{lem}\cite[Lemma 3.2]{CDJP}\label{3.2}
Let $\sigma$ contain only finitely many relation symbols, let $\mathcal{A}=\varprojlim_{i\in I}\mathcal{A}_i$ be a surjective inverse limit of
finite structures,
let $\mathcal{B}$ be a finite structure, and let $\varphi\colon\mathcal{A}\to\mathcal{B}$ be a $[$continuous$]$ homomorphism.
Then there exist $i\in I$ and a $[$continuous$]$ homomorphism $\psi\colon\mathcal{A}_i\to\mathcal{B}$ such that $\varphi=\psi\pi_i$.
\end{lem}


\section{Main results}

\noindent
In the proof of the following theorem, we use ideas from the proof of \cite[Theorem 4]{KNS4} as well as certain statements from the proof of \cite[Theorem 5.1]{B-I}.

\begin{thm}\label{T:1}
Let a class $\mathbf{A}=\{\mathcal{A}_F\mid F\in\mathcal{P}_{fin}(\omega)\}$ of structures
of a finite similarity type $\sigma$
be a finite $\mathrm{B}^\ast$-class with respect to a quasivariety $\mathbf{M}\subseteq\mathbf{K}(\sigma)$ and a variety $\mathbf{V}\subseteq\mathbf{K}(\sigma)$ which satisfies $(\mathrm{B}^\ast)$.
Then there are continuum many $Q$-universal subquasivarieties in $\mathbf{M}$ which are not profinite and have no finitely partitionable $\omega$-independent quasi-equational basis
relative to $\mathbf{M}$.
\end{thm}

\begin{proof}
By Remark \ref{R:1} and Corollary \ref{C:B*}, $\mathbf{A}$ satisfies $(\mathrm{B}^\ast_\Theta)$ for some $\Theta$ with
\begin{align*}
&\Theta=\bigl(\Delta(\Theta),\Gamma(\Theta),\Omega(\Theta)\bigr),\quad\text{where}\\
&\Delta(\Theta)=\bigl\{(X_F,\Delta_F,\gamma_F)\mid F\in\mathcal{P}_{fin}(\omega){\setminus}\{\varnothing\}\bigr\};\\
&\Gamma(\Theta)=\{\gamma^F_G\mid F,G\in\mathcal{P}_{fin}(\omega){\setminus}\{\varnothing\},\ G\subseteq F\};\\
&\Omega(\Theta)=\{f^F_G\mid F,G\in\mathcal{P}_{fin}(\omega){\setminus}\{\varnothing\},\ G\subseteq F\}.
\end{align*}
Following the proof of \cite[Theorem 5.1]{B-I}, we fix one such $\Theta$.
Moreover, we fix an infinite set
$I\subseteq\omega$  such that $\omega{\setminus}I\ne\varnothing$ and assume that
$I=\{i_n\mid n<\omega\}$, where $i_n\leqslant i_m$ if and only if $n\leqslant m$ for all $m,n<\omega$.
As in the proof of \cite[Theorem 4]{KNS4}, we also fix
an element $k\in\omega{\setminus}I$ and we put
\begin{align*}
&F_n=\{i_0,\ldots,i_n\}\quad\text{for all}\ n<\omega;\\
&\mathcal{A}=\mathcal{A}_{\{k\}},\quad\mathcal{A}_n=\mathcal{A}_{F_n}\quad\text{and}\quad\mathcal{B}_n=\mathcal{A}_{\{k\}\cup F_n}
\quad\text{for all}\ n<\omega.
\end{align*}
As in the proof of Theorem 5.1 in \cite{B-I}, we consider the following 
sentence $\varphi^I_F$ which is equivalent to a finite set of quasi-identities:
\[
\forall\,\overline{x}\ \ \&\Delta_F(\overline{x})\ \longrightarrow\ \&\Delta^F_{F\cap I}(\overline{x}),
\]
where $\mathcal{A}_F\cong\mathcal{F}_\mathbf{K}(X_F,\Delta_F)$ with a canonical interpretation $\gamma_F$ for all $F\in\mathcal{P}_{fin}(\omega)$. We put also
\[
\Phi_I=\bigl\{\varphi^I_F\mid F\in\mathcal{P}_{fin}(\omega),\ F\cap I\ne\varnothing\bigr\}\quad
\text{and}\quad\mathbf{K}_I=\operatorname{Mod}\Phi_I\cap\mathbf{M}.
\]
By \cite[Theorem 5.1]{B-I}, $\mathbf{K}_I$ is $Q$-universal and has no finitely partitionable $\omega$-indepen\-dent quasi-equational basis relative to $\mathbf{M}$.
The following statement is the content of Claim 5.2 from the proof of \cite[Theorem 5.1]{B-I}.

\setcounter{claim}{0}
\begin{claim}\label{C:1}
For each $F\in\mathcal{P}_{fin}(\omega)$,
$\mathcal{A}_F\in\mathbf{K}_I$ if and only if $F\subseteq I$.
\end{claim}

\noindent
We construct an inverse spectrum $\Lambda=\langle\omega,\mathcal{B}_j,\pi_{ij}\rangle$ as in the proof of \cite[Theorem 4]{KNS4}.
By Lemma \ref{L:B}(ii), for each $n<\omega$, there is a surjective homomorphism $\pi_{n,n+1}\colon\mathcal{B}_{n+1}\to\mathcal{B}_n$.
We put for all $i\leqslant n<\omega$:
\[
\pi_{in}=
\begin{cases}
\mathrm{id}_{\mathcal{B}_i},\quad&\text{if}\ i=n;\\
\pi_{i,{i+1}}\ldots\pi_{{n-1},n},\quad&\text{if}\ i<n.
\end{cases}
\]
We obtain therefore 

\begin{claim}\label{C:5}
$\Lambda=\langle\omega,\mathcal{B}_j,\pi_{ij}\rangle$ is a surjective inverse limit.
\end{claim}

\noindent
We put $\mathcal{B}=\varprojlim\Lambda$.
Since $B_n$ is a finite set for all $n<\omega$, we conclude that $B\ne\varnothing$.
Moreover, $\mathcal{B}\in\mathbf{S}\mathbf{P}(\mathbf{A})\subseteq\mathbf{Q}(\mathbf{A})\subseteq\mathbf{M}$.

\begin{claim}\label{C:2}
$\mathcal{B}$ is an infinite structure and $\mathcal{B}\in\mathbf{K}_I$.
\end{claim}

\begin{scproof} 
We show first that $\mathcal{B}\in\mathbf{K}_I$.
As $\mathcal{B}\in\mathbf{M}$, it suffices to establish that $\mathcal{B}\models\Phi_I$.
We consider therefore an arbitrary sentence $\varphi_F^I\in\Phi_I$; then $F\cap I\ne\varnothing$.
Let $\gamma\colon X_F\to B$ be an interpretation such that
$\mathcal{B}\models\Delta_F[\gamma]$.
This means that there is a homomorphism
$f\colon\mathcal{A}_F\to\mathcal{B}$ such that $f\bigl(\gamma_F(x)\bigr)=\gamma(x)$ for each $x\in X_F$.
For each $n<\omega$, let $f_n=\pi_nf$, where
$\pi_n\colon\mathcal{B}\to\mathcal{B}_n$ denotes the canonical projection from $\mathcal{B}$ onto $\mathcal{B}_n$.
Two cases are possible.

\Case{1}: $f(\mathcal{A}_F)\in\mathbf{V}$.

\Case{2}: $f(\mathcal{A}_F)\notin\mathbf{V}$. In this case, we obtain by Lemma~\ref{L:B}(iii) that $f_n(\mathcal{A}_F)\cong\mathcal{A}_G\leq\mathcal{B}_n$ for some nonempty set $G\subseteq F$.
Therefore, $(\mathrm{B}^\ast_2)$ yields the equality $G=\{k\}\cup F_n$.
This implies that $f_n(\mathcal{A}_F)\cong\mathcal{B}_n$ whence $|B_n|\leqslant|A_F|$ in this case.

\vskip3mm
\noindent
Since $\mathcal{A}_F$ is a finite structure and $|B_n| <|B_{n+1}|$ for all $n<\omega$ by Lemma \ref{L:B}(ii), there is $s<\omega$ such that $|A_F|<|B_n|$
for all $n\geqslant s$.
Therefore, $f_n(\mathcal{A}_F)\in\mathbf{V}$ for all $n\geqslant s$.
Let $J_s=\{n<\omega\mid n\geqslant s\}$; we have then
\begin{align*}
&\mathcal{B}\cong\varprojlim\langle J_s,\mathcal{B}_j,\pi_{ij}\rangle\leq_s\prod_{n\in J_s}\mathcal{B}_n;\\
&f(\mathcal{A}_F)\leq_s\prod_{n\in J_s}f_n(\mathcal{B})\in\mathbf{V}.
\end{align*}
This yields that the substructure $f(\mathcal{A}_F)$ of $\mathcal{B}$ generated by the set $\{\gamma(x)\mid x\in F\}$ belongs to $\mathbf{V}$.
Using our assumption that $F\cap I\ne\varnothing$ and applying $(\mathrm{B}^\ast_\Theta)$, we obtain that there is a homomorphism
$g\colon\mathcal{A}_{F\cap I}\to\mathcal{B}$ such that $g\bigl(\gamma^F_{F\cap I}(x)\bigr)=f\bigl(\gamma^F(x)\bigr)=\gamma(x)$ for all $x\in X_F$.
As $\mathcal{A}_{F\cap I}\models\Delta_{F\cap I}^F[\gamma^F_{F\cap I}]$, we conclude that $\mathcal{B}\models\Delta_{F\cap I}^F[\gamma]$ and $\mathcal{B}\models\varphi^I_F$ therefore.

Finally, if $|B|<\omega$ then by Lemma \ref{3.2}, there is $n<\omega$ such that $|B|\leqslant|B_n|$.
Hence, $|B|\leqslant|B_n|\leqslant|B_m|$ for all $m\geqslant n$ by Lemma \ref{L:B}(ii).
As $\pi_n$ is a homomorphism from $\mathcal{B}$ onto $\mathcal{B}$, we have that
$|B_m|\leqslant|B|\leqslant|B_m|$ and $|B_m|=|B|$ for all $m\geqslant n$ which is impossible as $|\sigma^P|<\omega$.
\end{scproof}

\noindent
According to Claim \ref{C:1}, we have $\mathcal{A}\notin\mathbf{K}_I$, $\mathcal{A}_n\in\mathbf{K}_I$ and $\mathcal{B}_n\notin\mathbf{K}_I$ for all $n<\omega$.
By $(\mathrm{B}_1)$, $\mathcal{B}_n\in\mathbf{Q}(\mathcal{A},\mathcal{A}_n)$.
As $\mathcal{B}_n\notin\mathbf{V}$ and $\mathcal{A}$ and $\mathcal{A}_n$ are finite structures of finite type,
we conclude by $(\mathrm{B}_2^\ast)$ that $\mathcal{B}_n$ is a subdirect product of structures isomorphic to $\mathcal{A}$ or to $\mathcal{A}_n$.
Lemma \ref{L:H}(iv) yields that $\mathcal{B}_n\leq_s\mathcal{A}\times\mathcal{A}_n$.
For each $n<\omega$, let
\[
\alpha_n\colon\mathcal{B}_n\to\mathcal{A}\quad\text{and}\quad\beta_n\colon\mathcal{B}_n\to\mathcal{A}_n
\]
denote the canonical surjective homomorphisms corresponding to these subdirect decompositions.
Since $\mathcal{B}_n\notin\mathbf{K}_I$ and $\mathcal{A}_n\in\mathbf{K}_I$,
we obtain that $\ker\beta_n\neq 0_{\mathcal{B}_n}$ for each $n<\omega$.
This means that, for each $n<\omega$,
there exists a relation symbol $p_n\in\sigma^P\cup\{=\}$ of arity $m_n$ and elements $b^n_1,\ldots,b^n_{m_n}\in B_n$ such that
\[
\mathcal{B}_n\not\models p_n(b^n_1,\ldots,b^n_{m_n})\quad\text{and}\quad\mathcal{A}_n\models p_n\bigl(\beta_n(b^n_1),\ldots,\beta_n(b^n_{m_n})\bigr).
\]
This yields that $\mathcal{A}\not\models p_n\bigl(\alpha_n(b^n_1),\ldots,\alpha_n(b^n_{m_n})\bigr)$.
Since $\sigma^P\cup\{=\}$ and $A$ are finite sets, there is an infinite set $J\subseteq\omega$ and a relation symbol
$p\in\sigma^P\cup\{=\}$ of arity $m$ such that $p_n=p$ and $m_n=m$ for all $n\in J$.

According to Lemma \ref{L:H}(ii), for each $n<\omega$, there is a surjective homomorphism $\varphi_{n,n+1}\colon\mathcal{A}_{n+1}\to\mathcal{A}_n$ such that
$\beta_n\pi_{n,n+1}=\varphi_{n,n+1}\beta_{n+1}$.
According to Lemma \ref{L:H}(iii), for each $n<\omega$, there is an isomorphism $\psi_{n,n+1}\colon\mathcal{A}\to\mathcal{A}$
such that $\psi_{n,n+1}\alpha_n\pi_{n,n+1}=\alpha_{n+1}$, see Figure \ref{F:1}.
We put for all $i\leqslant n<\omega$:
\[
\varphi_{in}=
\begin{cases}
\mathrm{id}_{\mathcal{A}_i},\ &\text{if}\ i=n;\\
\varphi_{i,{i+1}}\ldots\varphi_{{n-1},n},\ &\text{if}\ i<n.
\end{cases}
\qquad
\psi_{in}=
\begin{cases}
\mathrm{id}_{\mathcal{A}},\ &\text{if}\ i=n;\\
\psi_{{n-1},n}\ldots\psi_{i,{i+1}},\ &\text{if}\ i<n.
\end{cases}
\]
The for all $i\leqslant n<\omega$, we have
\[
\beta_i\pi_{in}=\varphi_{in}\beta_n,\quad
\psi_{in}\alpha_i\pi_{in}=\alpha_n;
\]
see Figure \ref{F:1}.
The proof of the following claim repeats the proof of Claim 4 from the proof of \cite[Theorem 4]{KNS4}.

\begin{figure}
\begin{center}
\setlength{\unitlength}{1mm}
\begin{picture}(120,50)
\put(0,20){$\mathcal{B}_0$}
\put(18,21){\vector(-1,0){12}}
\put(10,23){$\pi_{01}$}
\put(2,25){\vector(0,1){12}}
\put(3,30){$\beta_0$}
\put(20,20){$\mathcal{B}_1$}
\put(38,21){\vector(-1,0){12}}
\put(30,23){$\pi_{12}$}
\put(22,25){\vector(0,1){12}}
\put(23,30){$\beta_1$}
\put(40,20){$\mathcal{B}_2$}
\put(58,21){\vector(-1,0){12}}
\put(50,23){$\pi_{23}$}
\put(42,25){\vector(0,1){12}}
\put(43,30){$\beta_2$}

\put(62,21){\circle*{1}}
\put(64,21){\circle*{1}}
\put(66,21){\circle*{1}}

\put(62,31){\circle*{1}}
\put(64,31){\circle*{1}}
\put(66,31){\circle*{1}}

\put(82,21){\vector(-1,0){12}}
\put(72,23){$\pi_{{n-1},n}$}
\put(85,20){$\mathcal{B}_n$}
\put(105,21){\vector(-1,0){12}}
\put(95,23){$\pi_{n,{n+1}}$}
\put(87,25){\vector(0,1){12}}
\put(88,30){$\beta_n$}

\put(109,21){\circle*{1}}
\put(111,21){\circle*{1}}
\put(113,21){\circle*{1}}

\put(109,31){\circle*{1}}
\put(111,31){\circle*{1}}
\put(113,31){\circle*{1}}

%%%%%%%%%%%%%%%%%%%%%%%

\put(0,0){$\mathcal{A}$}
\put(6,1){\vector(1,0){12}}
\put(10,3){$\psi_{01}$}
\put(2,17){\vector(0,-1){12}}
\put(3,10){$\alpha_0$}
\put(20,0){$\mathcal{A}$}
\put(26,1){\vector(1,0){12}}
\put(30,3){$\psi_{12}$}
\put(22,17){\vector(0,-1){12}}
\put(23,10){$\alpha_1$}
\put(40,0){$\mathcal{A}$}
\put(46,1){\vector(1,0){12}}
\put(50,3){$\psi_{23}$}
\put(42,17){\vector(0,-1){12}}
\put(43,10){$\alpha_2$}

\put(62,1){\circle*{1}}
\put(64,1){\circle*{1}}
\put(66,1){\circle*{1}}

\put(62,11){\circle*{1}}
\put(64,11){\circle*{1}}
\put(66,11){\circle*{1}}

\put(70,1){\vector(1,0){12}}
\put(72,3){$\psi_{{n-1},n}$}
\put(85,0){$\mathcal{A}$}
\put(93,1){\vector(1,0){12}}
\put(95,3){$\psi_{n,{n+1}}$}
\put(87,17){\vector(0,-1){12}}
\put(88,10){$\alpha_n$}

\put(109,1){\circle*{1}}
\put(111,1){\circle*{1}}
\put(113,1){\circle*{1}}

\put(109,11){\circle*{1}}
\put(111,11){\circle*{1}}
\put(113,11){\circle*{1}}

%%%%%%%%%%%%%%%%%%%%%%%%%%%


\put(0,40){$\mathcal{A}_0$}
\put(18,41){\vector(-1,0){12}}
\put(10,43){$\varphi_{01}$}
\put(20,40){$\mathcal{A}_1$}
\put(38,41){\vector(-1,0){12}}
\put(30,43){$\varphi_{12}$}
\put(40,40){$\mathcal{A}_2$}
\put(58,41){\vector(-1,0){12}}
\put(50,43){$\varphi_{23}$}

\put(62,41){\circle*{1}}
\put(64,41){\circle*{1}}
\put(66,41){\circle*{1}}

\put(82,41){\vector(-1,0){12}}
\put(72,43){$\varphi_{{n-1},n}$}
\put(85,40){$\mathcal{A}_n$}
\put(105,41){\vector(-1,0){12}}
\put(95,43){$\varphi_{n,{n+1}}$}

\put(109,41){\circle*{1}}
\put(111,41){\circle*{1}}
\put(113,41){\circle*{1}}

\end{picture}
\end{center}
\caption{\ }\label{F:1}
\end{figure}


\begin{claim}\label{C:3}
If $(b_1,\ldots,b_m)\in\ker\beta_n(p){\setminus}\ker\alpha_n(p)$ for some $n<\omega$ then
\[
\bigl(\pi_{tn}(b_1),\ldots,\pi_{tn}(b_m)\bigr)\in\ker\beta_t(p){\setminus}\ker\alpha_t(p)
\]
for all $t\leqslant n$.
\end{claim}

\noindent
Since the set $J\subseteq\omega$ is infinite,
it follows from Claim \ref{C:3} that for each $n<\omega$, there are elements $b^n_1,\ldots,b^n_m\in B_n$ such that
\begin{align*}
&\mathcal{B}_n\not\models p(b^n_1,\ldots,b^n_m);\\
&\mathcal{A}_n\models p\bigl(\beta_n(b^n_1),\ldots,\beta_n(b^n_m)\bigr)\quad\text{and thus,}\ (b^n_1,\ldots,b^n_m)\in\ker\beta_n(p);\\
&\mathcal{A}\not\models p\bigl(\alpha_n(b^n_1),\ldots,\alpha_n(b^n_m)\bigr)\quad\text{and thus,}\ (b^n_1,\ldots,b^n_m)\notin\ker\alpha_n(p).
\end{align*}


\begin{claim}\label{C:4}
There are elements $c_1,\ldots,c_m\in B$ such that $\mathcal{B}\not\models p(c_1,\ldots,c_m)$ and,
for each homomorphism $f\colon\mathcal{B}\to\mathcal{D}$ where $\mathcal{D}\in\mathbf{K}_I$ is a finite structure,
$\mathcal{D}\models p\bigl(f(c_1),\ldots,f(c_m)\bigr)$.
\end{claim}

\begin{scproof}
Exactly as in the proof of Claim 5 from the proof of \cite[Theorem 4]{KNS4}, we can construct elements
$c_1,\ldots,c_m\in B$ such that
\[
\bigl(\pi_n(c_1),\ldots,\pi_n(c_m)\bigr)\in\ker\beta_n(p){\setminus}\ker\alpha_n(p)\quad\text{for all}\ n<\omega.
\]
As $\bigl(\pi_n(c_1),\ldots,\pi_n(c_m)\bigr)\notin\ker\alpha_n(p)$ for all $n<\omega$, we have $\mathcal{B}\not\models p(c_1,\ldots,c_m)$.

Consider an arbitrary homomorphism $f\colon\mathcal{B}\to\mathcal{D}$ with $\mathcal{D}\in\mathbf{K}_I$ being a finite structure.
By Lemma \ref{3.2}, there is $n<\omega$ and a homomorphism $g\colon\mathcal{B}_n\to\mathcal{D}$ such that
$f=g\pi_n$. Two cases are possible.

\Case{1}: $g(\mathcal{B}_n)\in\mathbf{V}$. As $\mathbf{A}$ satisfies $(\mathrm{B}^\ast_\Theta)$, there is a homomorphism
$h\colon\mathcal{A}_n\to\mathcal{D}$ such that
$h\bigl(\gamma^{\{k\}\cup F_n}_{F_n}(x)\bigr)=f\bigl(\gamma^{\{k\}\cup F_n}(x)\bigr)$ for all $x\in X_{\{k\}\cup F_n}$.
As the set $X_{\{k\}\cup F_n}$ generates $\mathcal{B}_n$, we conclude that
$g=hf^{\{k\}\cup F_n}_{F_n}=h\beta_n$ whence $f=g\pi_n=h\beta_n\pi_n$.

\Case{2}: $g(\mathcal{B}_n)\notin\mathbf{V}$.
By Lemma \ref{L:B}(iii), we conclude that $g(\mathcal{B}_n)\cong\mathcal{A}_G\leq\mathcal{D}\in\mathbf{K}_I$
for some $G\subseteq\{k\}\cup F_n$.
Thus, $\mathcal{A}_G\in\mathbf{K}_I$ whence $G\subseteq(\{k\}\cup F_n)\cap I=F_n$ by Claim \ref{C:1}.
By Lemma \ref{L:H}(ii), there is a homomorphism $h\colon\mathcal{A}_n\to\mathcal{D}$ with
$g=h\beta_n$; therefore, $f=h\beta_n\pi_n$.

\vskip3mm
\noindent
In both cases, $f=h\beta_n\pi_n$ for some homomorphism $h\colon\mathcal{A}_n\to\mathcal{D}$.
This implies, in particular, that $\ker\beta_n\pi_n\leq\ker f$.
Since $\bigl(\pi_n(c_1),\ldots,\pi_n(c_m)\bigr)\in\ker\beta_n(p)$, we conclude that $(c_1,\ldots,c_m)\in\ker\beta_n\pi_n(p)\subseteq\ker f(p)$.
Therefore, $\mathcal{D}\models p\bigl(f(c_1),\ldots,f(c_m)\bigr)$.
\end{scproof}

\noindent
By Claims \ref{C:2} and \ref{C:4}, the profinite structure $\mathcal{B}\in\mathbf{K}_I$ does not embed into a Cartesian product of finite structures from
$\mathbf{K}_I$, whence it is not profinite with respect to
$\mathbf{K}_I$. Hence, the quasivariety $\mathbf{K}_I$ is not profinite.

If $I,J\subseteq\omega$ are such that $I\nsubseteq J$ then there is $i\in I{\setminus}J$.
According to Claim \ref{C:1}, $\mathcal{A}_i\in\mathbf{K}_I{\setminus}\mathbf{K}_J$,
whence $\mathbf{K}_I\nsubseteq\mathbf{K}_J$.
It remains to recall that there are continuum many infinite proper subsets $I\subset\omega$.
\end{proof}

\noindent
In the proof of the following theorem, we use ideas from the proof of \cite[Theorem 5]{KNS4} as well as certain statements from the proof of \cite[Theorem 2.1]{B-II}.

\begin{thm}\label{T:2}
Let a class $\mathbf{A}=\{\mathcal{A}_F\mid F\in\mathcal{P}_{fin}(\omega)\}$ of structures
of a finite similarity type $\sigma$
be a finite $\mathrm{B}^\ast$-class with respect to a quasivariety $\mathbf{M}\subseteq\mathbf{K}(\sigma)$ and a variety $\mathbf{V}\subseteq\mathbf{K}(\sigma)$ which satisfies $(\mathrm{B}^\ast)$.
Then there are continuum many $Q$-universal subquasivarieties in $\mathbf{M}$ which are not profinite and have a finitely partitionable $\omega$-independent quasi-equational basis
relative to $\mathbf{M}$.
\end{thm}

\begin{proof}
Again, we fix $\Theta$ and an infinite set
$I\subseteq\omega$ and assume that
$I=\{i_n\mid n<\omega\}$, where $i_n\leqslant i_m$ if and only if $n\leqslant m$ for all $m,n<\omega$.
As in the proof of \cite[Theorem 5]{KNS4}, we also put
\begin{align*}
&F_\bot=\{i_1\}\quad\text{and}\quad\mathcal{B}_\bot=\mathcal{A}_{F_\bot}
&F_n=\{i_0,\ldots,i_n\}\quad\text{and}\quad\mathcal{B}_n=\mathcal{A}_{F_n}\quad\text{for all}\ n<\omega.
\end{align*}
As in the proof of \cite[Theorem 2.1]{B-I}, see also Theorem \ref{T:B}, we consider for each $m<\omega$ the following 
sentence which is equivalent to a finite set of quasi-identities:
\[
\forall\,\overline{x}\quad\&\Delta_{F_{m+1}}(\overline{x})\ \longrightarrow\ \&\Delta^{F_{m+1}}_{F_m}(\overline{x}).
\]
We denote this sentence by $\xi_m$.

We have $F_\bot=\{i_0\}\subseteq\{i_0,i_1\}=F_1$.
By $\xi_\bot$, we denote the following sentence which is also equivalent to a finite set of quasi-identities:
\[
\forall\,\overline{x}\quad\&\Delta_{F_1}(\overline{x})\ \longrightarrow\ \&\Delta^{F_1}_{F_\bot}(\overline{x}).
\]
We put
\[
\Xi_I=\{\xi_m\mid m<\omega\}\cup\{\xi_\bot\}\quad
\text{and}\quad\mathbf{M}_I=\operatorname{Mod}\Xi_I\cap\mathbf{M}.
\]
By Claims 2.5 and 2.7 from the proof of \cite[Theorem 2.1]{B-I}, see also Theorem \ref{T:B}, we obtain the following

\setcounter{claim}{0}
\begin{claim}\label{C:21}
$\mathbf{M}_I$ is a $Q$-universal quasivariety which consists of all structures $\mathcal{A}\in\mathbf{M}$ with the following property:
\[
\text{if}\ \mathcal{A}_F\in\mathbf{S}(\mathcal{A})\ \text{for some nonempty}\  F\in\mathcal{P}_{fin}(\omega)\ \text{then}\ F\nsubseteq I.
\]
Moreover, $\Xi_I$ is a finitely partitionable $\omega$-independent quasi-equational basis of $\mathbf{M}_I$ relative to $\mathbf{M}$.
\end{claim}

\noindent
Let $\omega^\prime=\omega\cup\{\bot\}$ where $\bot\leqslant n$ for all $n$ such that $0<n<\omega$.
It is clear that $\omega^\prime$ is an up-directed set with respect to $\leqslant$.
We construct an inverse spectrum $\Lambda=\langle\omega^\prime,\mathcal{B}_j,\pi_{ij}\rangle$ as follows.
According to Lemma \ref{L:B}(ii), for each $n<\omega$, there is a surjective homomorphism $\pi_{n,n+1}\colon\mathcal{B}_{n+1}\to\mathcal{B}_n$.
We put for all $i\leqslant n<\omega$:
\[
\pi_{in}=
\begin{cases}
\mathrm{id}_{\mathcal{B}_i},\quad&\text{if}\ i=n;\\
\pi_{i,{i+1}}\ldots\pi_{{n-1},n},\quad&\text{if}\ i<n.
\end{cases}
\]
By Lemma \ref{L:B}(ii), there is a surjective homomorphism $\pi_{\bot,1}\colon\mathcal{B}_1\to\mathcal{B}_\bot$.
We put $\pi_{\bot\bot}=\mathrm{id}_{\mathcal{B}_\bot}$ and
$\pi_{\bot n}=\pi_{\bot,1}\ldots\pi_{{n-1},n}$ for all $n$ such that $0<n<\omega$:

The next statement follows immediately. 

\begin{claim}\label{C:22}
$\Lambda=\langle\omega^\prime,\mathcal{B}_j,\pi_{ij}\rangle$ is a surjective inverse limit.
\end{claim}

\noindent
We put $\mathcal{B}=\varprojlim\Lambda$.
Since $B_n$ is a finite set for all $n\in\omega^\prime$, we conclude that $B\ne\varnothing$.
Moreover, $\mathcal{B}\in\mathbf{S}\mathbf{P}(\mathbf{A})\subseteq\mathbf{Q}(\mathbf{A})\subseteq\mathbf{M}$.

\begin{claim}\label{C:23}
$\mathcal{B}$ is an infinite structure and $\mathcal{B}\in\mathbf{M}_I$.
\end{claim}

\begin{scproof}
Exactly as in the proof of Claim \ref{C:2} from the proof of Theorem \ref{T:1}, one establishes that $\mathcal{B}$ is an infinite structure.
We show now that $\mathcal{B}\models\Xi_I$.
Assume that $m<\omega$ and that an interpretation
$\gamma\colon X_{F_{m+1}}\to B$ is such that
$\mathcal{B}\models\Delta_{F_{m+1}}[\gamma]$.
This means that there is a homomorphism
$f\colon\mathcal{B}_{m+1}\to\mathcal{B}$ such that $f\bigl(\gamma_{F_{m+1}}(x)\bigr)=\gamma(x)$ for each $x\in X_{F_{m+1}}$.
For each $n<\in\omega^\prime$, let $f_n=\pi_nf$ where $\pi_n\colon\mathcal{B}\to\mathcal{B}_n$ is a canonical projection.
Two cases are possible.

\Case{1}: $f_n(\mathcal{B}_{m+1})\in\mathbf{V}$.

\Case{2}: $f_n(\mathcal{B}_{m+1})\notin\mathbf{V}$.
In this case, we have by Lemma \ref{L:B}(iii) that $f_n(\mathcal{B}_{m+1})\cong\mathcal{A}_G\leq\mathcal{B}_n$ for some nonempty set $G\subseteq F_m$.
Therefore, we get by $(\mathrm{B}_2^\ast)$ that $G=F_n$.
This implies that $f_n(\mathcal{B}_{m+1})\cong\mathcal{B}_n$ whence $|B_n|\leqslant|B_{m+1}|$ in this case.

\vskip3mm
\noindent
Since $\mathcal{B}_{m+1}$ is a finite structure and $|B_n| <|B_{n+1}|$ for all $n<\omega$ by Lemma \ref{L:B}(ii), there is $s<\omega$ such that $|B_{m+1}|<|B_n|$
for all $n\geqslant s$.
Therefore, $f_n(\mathcal{B}_{m+1})\in\mathbf{V}$ for all $n\geqslant s$.
Let $J_s=\{n<\omega\mid n\geqslant s\}$; we have then
\begin{align*}
&\mathcal{B}\cong\varprojlim\langle J_s,\mathcal{B}_j,\pi_{ij}\rangle\leq_s\prod_{n\in J_s}\mathcal{B}_n;\\
&f(\mathcal{B}_{m+1})\leq_s\prod_{n\in J_s}f_n(\mathcal{B})\in\mathbf{V}.
\end{align*}
This yields that the substructure $f(\mathcal{B}_{m+1})$ of $\mathcal{B}$ generated by the set $\{\gamma(x)\mid x\in F_{m+1}\}$ belongs to $\mathbf{V}$.
As $F_m\ne\varnothing$ and $F_\bot\ne\varnothing$, we apply $(\mathrm{B}^\ast_\Theta)$ and obtain that there is a homomorphism
$g\colon\mathcal{B}_m\to\mathcal{B}$ such that $g\bigl(\gamma^{F_{m+1}}_{F_m}(x)\bigr)=f\bigl(\gamma^{F_{m+1}}(x)\bigr)=\gamma(x)$ for all $x\in X_{F_{m+1}}$,
as well as there is a homomorphism
$g\colon\mathcal{B}_\bot\to\mathcal{B}$ such that $g\bigl(\gamma^{F_1}_{F_\bot}(x)\bigr)=f\bigl(\gamma^{F_1}(x)\bigr)=\gamma(x)$ for all $x\in X_{F_1}$
whenever $m=0$.
As $\mathcal{B}_m\models\Delta_{F_m}^{F_{m+1}}[\gamma^{F_{m+1}}_{F_m}]$ and
$\mathcal{B}_\bot\models\Delta_{F_\bot}^{F_1}[\gamma^{F_1}_{F_\bot}]$,
we conclude that $\mathcal{B}\models\Delta_{F_m}^{F_{m+1}}[\gamma]$ and $\mathcal{B}\models\Delta_{F_\bot}^{F_1}[\gamma]$
whenever $m=0$.
This implies that $\mathcal{B}\models\xi_m$ for all $m<\omega$ and $\mathcal{B}\models\xi_\bot$.
\end{scproof}

\begin{claim}\label{C:24}
$\mathcal{B}$ is not profinite with respect to $\mathbf{M}_I$.
\end{claim}

\begin{scproof}
Consider an arbitrary homomorphism $f\colon\mathcal{B}\to\mathcal{D}$, where $\mathcal{D}\in\mathbf{M}_I$ is a finite structure.
By Lemma \ref{3.2}, there is $n\in\omega^\prime$ and a homomorphism $g\colon\mathcal{B}_n\to\mathcal{D}$ such that
$f=g\pi_n$. 
We have by Lemma \ref{L:B}(iii) and $(\mathrm{B}_2^\ast)$ that either $g(\mathcal{B}_n)\in\mathbf{V}$ or $g(\mathcal{B}_n)\cong\mathcal{A}_G\leq\mathcal{D}\in\mathbf{M}_I$
for some nonempty finite set $G$ such that $G\subseteq F_n\subseteq I$. In the second case, we obtain a contradiction with Claim \ref{C:21}.
This contradiction shows that $f(\mathcal{B})\cong g(\mathcal{B}_n)\in\mathbf{V}$.

If $\mathcal{B}$ were profinite with respect to $\mathbf{M}_I$ then $\mathcal{B}\leq_s\prod_{t\in T}\mathcal{D}_t$,
where $\mathcal{D}_t\in\mathcal{M}_I$ is a finite structure for each $t\in T$.
According to the above argument $\mathcal{D}_t\in\mathbf{V}$ for each $t\in T$ whence $\mathcal{B}\in\mathbf{V}$.
As $\mathbf{V}$ is a variety and $\mathcal{B}_n\in\mathbf{H}(\mathcal{B})$ for all $n\in\omega^\prime$, we conclude that
$\mathcal{B}_n\in\mathbf{V}$ for all $n\in\omega^\prime$ which contradicts Lemma \ref{L:B}(i).
\end{scproof}

\noindent
By Claims \ref{C:23} and \ref{C:24}, the profinite structure $\mathcal{B}\in\mathbf{M}_I$ is not profinite with respect to
$\mathbf{M}_I$, whence the quasivariety $\mathbf{M}_I$ is not profinite.

If $I,J\subseteq\omega$ are such that $I\nsubseteq J$ then there is $i\in I{\setminus}J$.
According to Claim \ref{C:21}, $\mathcal{A}_i\in\mathbf{M}_J{\setminus}\mathbf{M}_I$,
whence $\mathbf{M}_J\nsubseteq\mathbf{M}_I$.
The fact that there are continuum many infinite proper subsets in $\omega$ completes the proof.
\end{proof}


\section{Applications}\label{A}

\noindent
We present in this section some applications of Theorems \ref{T:1} and \ref{T:2} in a compact way. For more details, we refer to \cite{B-I,B-II}.

\begin{remark}
As each $\mathrm{B}$-class is a $\mathrm{B}^\ast$-class, \cite[Theorem 4]{KNS4} is a corollary of our Theorem \ref{T:1} while \cite[Theorem 5]{KNS4} is a corollary of Theorem \ref{T:2}.
In particular, all the statements of \cite[Corollary 1]{KNS4} follow from Theorems \ref{T:1}--\ref{T:2} by Corollary \ref{C:b-b}.
\end{remark}

\noindent
For the definition of relatively full embeddings and almost $ff$-universal quasivariety, we refer to A. Pultr and V. Trnkov\'{a} \cite{PT} and to V. Koubek and J. Sichler \cite{KS2,KS1}.

\begin{cor}\label{C:alU}
If $\mathbf{K}\subseteq\mathbf{K}(\sigma)$ is an almost ${f}{f}$-universal quasivariety of a finite type
$\sigma$, then there is a quasivariety $\mathbf{K}^\prime\subseteq\mathbf{K}$ which contains continuum many $Q$-universal non-profinite quasivarieties $\mathbf{M}\subseteq\mathbf{K}^\prime$ with one of the following properties:
\begin{enumerate}
\item
$\mathbf{M}$ has a finitely partitionable $\omega$-independent quasi-equational basis relative to some subquasivariety $\mathbf{K}^\prime$;
\item
$\mathbf{M}$ has no finitely partitionable $\omega$-independent quasi-equational basis relative to some subquasivariety $\mathbf{K}^\prime$.
\end{enumerate}
\end{cor}

\begin{proof}
By Theorem 6.2 and Corollary 6.11 of \cite{B-I}, $\mathbf{K}$ contains a finite $\mathrm{B}^\ast$-class relative to some subquasivariety $\mathbf{K}^\prime\subseteq\mathbf{K}$ and some variety which satisfies the condition $(\mathrm{B}^\ast)$.
The desired statement follows from Theorems \ref{T:1}--\ref{T:2}.
\end{proof}

\noindent
The statement of the next corollary was established in \cite{DG-UA-3}. For the definition of the variety $\mathbf{Dm}$ of differential groupoids,
we refer to A.\,B. Romanowska and J.\,D.\,H. Smith \cite{RS}, see also \cite{DG-UA-1}--\cite{DG-UA-3}, \cite[Section 4]{B-I} and \cite{BNSZ}.

\begin{thm}\label{T:ib}
\cite[Theorem 2.2]{B-II}
Let a class $\mathbf{A}=\{\mathcal{A}_F\mid F\in\mathcal{P}_{fin}(\omega)\}$ of structures
of a finite similarity type $\sigma$
be a $\mathrm{B}^\ast$-class satisfying $(\mathrm{B}^\ast_5)$ and $(\mathrm{B}^\ast)$ with respect to a quasivariety $\mathbf{M}\subseteq\mathbf{K}(\sigma)$
and a variety $\mathbf{V}\subseteq\mathbf{K}(\sigma)$.
Assume that
\begin{equation}\label{E:1}
\Delta^{F\cup\{n\}}_F\ \text{is equivalent within}\ \mathbf{M}\ \text{to}\ \Delta_F\cup\{\varphi^F_n\}
\end{equation}
where $\varphi^F_n$ is an atomic formula for all finite nonempty sets $F\in\mathcal{P}_{fin}(\omega)$ and all $n<\omega$ such that $n\notin F$.
Then there are continuum many quasivarieties $\mathbf{K}\subseteq\mathbf{M}$ with the following properties:
\begin{enumerate}
\item
$\mathbf{K}$ has a finitely partitionable $\omega$-independent quasi-equational basis relative to $\mathbf{M}$;
\item
there is a set $\mathfrak{I}$ and quasivarieties $\mathbf{K}_I\subseteq\mathbf{M}$, $I\in\mathfrak{I}$, such that
$\mathbf{K}=\bigcap_{I\in\mathfrak{I}}\mathbf{K}_I$ and the quasivariety $\mathbf{K}_I$
has no finitely partitionable $\omega$-independent quasi-equational basis relative to $\mathbf{M}$, but has
an $\omega$-independent quasi-equational basis relative to $\mathbf{M}$ for all $I\in\mathfrak{I}$.
\end{enumerate}
If $\mathbf{A}$ is a finite $\mathrm{B}^\ast$-class then $\mathbf{K}$ above can be chosen $Q$-universal.
\end{thm}

\begin{cor}\label{P:DG}
The variety $\mathbf{Dm}$ contains
continuum many $Q$-universal non-profin\-ite subquasivarieties $\mathbf{M}\subseteq\mathbf{Dm}$ with one the following properties:
\begin{enumerate}
\item
$\mathbf{M}$ has an independent quasi-equational basis relative to $\mathbf{Dm}$;
\item
$\mathbf{M}$ has no independent quasi-equational basis relative to $\mathbf{Dm}$.
\end{enumerate}
\end{cor}

\begin{proof}
By \cite[Proposition 7.1]{B-I}, the variety $\mathbf{Dm}$ contains a finite $\mathrm{B}^\ast$-class satisfying
$(\mathrm{B}^\ast)$ with respect to $\mathbf{Dm}$ and a finitely generated variety $\mathbf{V}\subseteq\mathbf{Dm}$
for which the assumptions of Theorem \ref{T:ib} are satisfied; in particular, the condition (\ref{E:1}) is fulfilled.
The desired statement follows from Theorems \ref{T:1}--\ref{T:2} and Theorem \ref{T:ib}.
\end{proof}

\begin{cor}\label{P:UA}
Let $\mathbf{K}^\prime$ be one of the following classes:
\begin{enumerate}
\item[(a)]
the variety of lattices generated by the modular lattice $M_{3,3}$;
\item[(b)]
one of the infinitely many pairwise incomparable lattice varieties generated by a single finite
simple lattice and not containing the lattice $M_{3,3}$ constructed in \emph{\cite{KS16}};
\item[(c)]
any variety of $0$-lattices containing a finite lattice with more than two elements and
no prime ideal.
\end{enumerate}
Then there is a quasivariety $\mathbf{M}\subseteq\mathbf{K}^\prime$ which contains continuum many
$Q$-universal non-profinite subquasivarieties $\mathbf{K}\subseteq\mathbf{M}$ with one the following properties:
\begin{enumerate}
\item
$\mathbf{K}$ has a finitely partitionable $\omega$-independent quasi-equational basis relative to $\mathbf{M}$;
\item
$\mathbf{K}$ has no finitely partitionable $\omega$-independent quasi-equational basis relative to $\mathbf{M}$.
\end{enumerate}
\end{cor}

\begin{proof}
As it was established in the papers of V. Koubek and J. Sichler \cite{KS16}--\cite{KS21}, in each of the cases (a)--(c),
$\mathbf{K}^\prime$ is an almost ${f}{f}$-universal quasivariety.
The desired statement follows from Corollary \ref{C:alU}.
\end{proof}




\end{document}